Recent Advances in Quantitative Gait Analysis Using Wearable Sensors: A Review

Yonatan Hutabarat[®], Dai Owaki[®], *Member, IEEE*, and Mitsuhiro Hayashibe[®], *Senior Member, IEEE*

Abstract—The current gold standard for gait analysis involves performing the gait experiments in a laboratory environment with a constrained space. However, there is growing interest in using flexible, efficient, and inexpensive wearable sensors as tools to perform gait analysis. This review aimed to identify and summarize the current advances in wearable sensors for various aspects of gait analysis, such as the application of wearable gait analysis systems, sensor systems and their attachment locations, and the algorithms used for the analysis. The PRISMA guideline was adopted to find relevant studies from the period 2011 to 2020 from several scientific databases. A total of 76 articles were selected based on the inclusion and exclusion criteria. A wearable inertial measurement unit (IMU) attached to the lower limb region was found to be the most common approach for gait analysis.



Sensors Council

Temporal, spatial, and spatiotemporal features were the most common quantitative gait features extracted from the wearable sensors. The proposed frameworks showed varying performances, and an increased number of sensors did not necessarily improve the estimation performance metrics. A few studies have integrated various machine learning techniques for classification problems, correction algorithms, crosschecking functions, and scoring functions. Finally, this review paper discusses the challenges and future direction of the research on quantitative gait analysis.

Index Terms—Gait analysis, wearable sensors, inertial measurement units, machine learning, walking, clinical applications.

I. INTRODUCTION

G AIT is often interpreted as the manner of walking. Quantitative gait analysis is the in-depth analysis of the gait of a person based on various quantifiable parameters. Gait analysis has wide applications ranging from sports to clinical investigations. In sports, gait analysis is often utilized to assess the performance of athletes, prevent injuries, and provide a training guide [1], [2]. In clinical applications, gait analysis is performed to characterize certain gait pathologies, track rehabilitation progress, and evaluate the effectiveness of certain treatments [3]–[5]. In addition to these applications,

Manuscript received July 22, 2021; revised September 21, 2021; accepted October 4, 2021. Date of publication October 13, 2021; date of current version November 30, 2021. This work was supported by the Japan Society for the Promotion of Science (JSPS) KAKENHI under Grant JP18H01399. The associate editor coordinating the review of this article and approving it for publication was Dr. Shanhong Xia. (Corresponding author: Yonatan Hutabarat.)

Yonatan Hutabarat is with the Neuro-Robotics Laboratory, Graduate School of Biomedical Engineering, Tohoku University, Sendai 980-8579, Japan (e-mail: yonatancah@dc.tohoku.ac.jp).

Dai Owaki and Mitsuhiro Hayashibe are with the Neuro-Robotics Laboratory, Department of Robotics, Graduate School of Engineering, Tohoku University, Sendai 980-8579, Japan (e-mail: owaki@tohoku.ac.jp; hayashibe@tohoku.ac.jp).

Digital Object Identifier 10.1109/JSEN.2021.3119658

gait analysis can be used to predict the risk of falling in elderly subjects [6], [7].

Several approaches can be used to perform the gait analysis. In most clinical settings, a combination of observation and qualitative assessment by clinicians and self-reported assessments by patients is employed [8], [9]. Observations collected by clinicians such as doctors or physical therapists may provide some quantitative gait features such as distance covered, total walking time, gait speed, and cadence. Nevertheless, these are subject to inter-observer variability and human error.

Presently, it is very common to perform quantitative gait analysis in a laboratory environment by using a gold standard measurement such as the combination of motion capture and force plate systems. Motion capture enables precise tracking of the spatial information of human motion in 3D, whereas the force plates provide dynamic features such as ground reaction forces and moments. However, the use of these specialized instruments is limited to only a few clinics or research facilities and have limited capture volume; thus, they may not necessarily capture the natural gait of the subject [10]. Another concern is the considerable time required for preparing for the experiment, such as placing markers and conducting anthropometric measurements, which may not be convenient for patients participating in the study.

This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

The recent vast developments in sensor systems and computational methods have made it possible to assess the gait outside the laboratory through wearable sensor systems. In addition to gait analysis, wearable sensors have been investigated to predict several clinical conditions based on the resting heart rate, temperature, and other features [11], [12]. In this review, we focus on the recent advancements in wearable-based gait analysis, which has recently been widely adopted in various applications. The number of sensors used may depend on the subject of interest. For example, sensors attached to both legs are preferred for a highly impaired gait [13]. Wearable inertial sensor-based gait analysis also serves as a supplementary assessment of the physical function of patients undergoing total hip arthroplasty [14] and in clinical analysis of idiopathic normal pressure hydrocephalus [19]. Several studies have discussed the long-term assessment of gait using wearable sensors [15]. In clinical applications, it has been reported that there is a strong correlation between accelerometer-based motor fluctuation measurement and the gait item in the Unified Parkinson's Disease Rating Scale-Part III [16], which suggests that the accelerometer may be a useful tool for monitoring Parkinson's disease (PD) patients. Wearable inertial sensors have also been used in biofeedback systems for gait and balance training of PD subjects [17], [18]. Gait quality metrics were found to be robust in assessing the gait of two matched groups of multiple sclerosis patients at two separate locations with different experimental protocols [20]. In assessing gait ataxia in people with spinocerebellar degeneration, the accelerometer-based assessment is more responsive than the standard clinical scale for the assessment and rating of ataxia [21]. Nevertheless, the performance accuracy and types of analysis are dependent on many factors such as the types of sensors, number of sensors, their positions on the body, and the methods and algorithms used. Therefore, this study aimed to address the recent advances in wearable-based gait analysis based on the above mentioned factors. The overall scientific contributions of this review paper are as follows:

- It presents an overview of the recent state of the art in quantitative gait analysis using wearable sensors, which have been validated against the gold standard measurement or other measurement systems.
- It provides key insights into the existing wearabletechnology-based gait analysis in terms of the number and types of sensors, gait features, positioning of the sensors on the body, as well as the method and algorithms used in each study.
- It highlights the current applications of wearable technology based gait analysis for both general purposes such as sports and clinical purposes related to various pathological conditions.
- It discusses the issues, challenges, and future research direction for researchers and clinicians working in this field.

The remainder of this paper is organized as follows. Section II presents the methods for locating relevant papers on gait analysis, as well as the exclusion and inclusion criteria for the paper to be reviewed. Section III presents the results of the literature search, which are classified based on key topics such as type and number of sensors, application of wearable gait analysis, sensor positions on the body, and algorithms implemented to extract the gait features. Section IV discusses the findings, current challenges, and possible future research directions. Finally, Section V concludes the review.

II. METHODS

The method used to find relevant references containing state-of-the-art techniques followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guideline [22], which is commonly used for reporting systematic reviews. The search was conducted on several databases including PubMed, IEEE Xplore, Scopus, and ISI Web of Science for the period from 2011 until 2020. The time constraint was added to ensure that only recent advances in this topic were considered for the review. Search keywords and combinations were formulated, as shown in Fig. 1. In the case of abbreviated search terms such as IMU and GRF, the search was repeated with their expanded forms, i.e., "inertial measurement unit" and "ground reaction force," respectively. We limited the search to only the titles of the articles along with the time constraint from 2011 to 2020. Gait was a necessary word in the document title, followed by any combination of keywords, as depicted in Fig. 1.

The database search was performed on February 8, 2021. A total of 1550 records from all the databases were included in the selection process. These results were then imported into the Mendeley® Desktop reference management software (version 1.19.4, Elsevier). Duplicate records were automatically removed by the software if an exact match was detected. In all, 860 duplicates were identified, resulting in 690 records that needed to be screened.

The screening stage excluded certain studies on the topics depicted in Fig. 1 based on the title and abstract. Assistive walking technologies were not included in this study, as most of these placed the sensors on the walking device, such as on a cane or a walker. Studies using non-wearable sensors, such as studies employing cameras or Kinect or any combination of these systems, were excluded. Smartphone-based studies were excluded because they are not a fully dedicated system and are mostly used for activity recognition or human authentication as opposed to the detailed gait analysis targeted in this review. Other unrelated studies such as those on sensor development, sensor-to-body alignment, underwater gait, and person/gender recognition were also detected and excluded in the screening stage. At the end of the screening stage, 344 records were excluded based on the set criteria, leaving 356 articles for the next stage.

In the eligibility stage, the full-text articles were accessed to find relevant studies to review the state of the art. The inclusion criteria for full-text screening are shown in Fig. 1. We preferred studies that set a benchmark or compared their proposed framework/system with a gold standard system or other existing studies. Moreover, each of the studies addressed various gait features, as depicted in Fig. 1. Nevertheless, we found some studies that used sensors with cables tethered to a data processing unit such as a computer, and those studies were discarded. Another criterion was that the studies must



Fig. 1. Formulated search keywords on several scientific databases with the time constraint, followed by the PRISMA flowchart for inclusion process and exclusion criterion of the articles selected in this study. Asterisk (*) indicates wildcard for the search term.

address more than one gait feature. At the end of this stage, 76 articles were found eligible for a systematic review. The overall flowchart of the article selection process is depicted in Fig. 1, and the distribution of eligible articles is depicted in Fig. 2.

A. Existing Review

While doing this process, a total of 21 review papers were also identified regarding this topic. Among them, there are six review papers published in 2020, which are described briefly as the following. Kobsar *et al.* on [23] addressed a specific application of wearable inertial sensors for adults with osteoarthritis. Gondim *et al.* [24] discussed the use of the portable accelerometer to evaluate the gait of people with Parkinson's disease. Junior *et al.* [25] was focused on reviewing gait assessment in children. However, all of the above papers discussed a very specific subject and conditions. Díaz *et al.* [26] discussed a wider scope of gait that includes



Fig. 2. Distribution of the eligible articles based on year (bar chart) and publication types (pie chart).

balance and range of motion analysis using wearable sensors. Dasgupta *et al.* [27] was focused on acceleration gait measure for motor skill assessment. Hence, it was specific to motor skills instead of gait. Saboor *et al.* [28] discussed the latest trend on gait analysis using wearable sensors combined with machine learning-based methods. However, the study selected was constrained from 2015 onwards with limitations only to machine learning methods. Other notable review papers are from Tao *et al.* [29] in 2012 that discussed basic human gait analysis and sensors system, Lopez-Nava and Munoz-Melendez [30] in 2017 that focused on wearable inertial sensors for motion analysis, and Caldas *et al.* [31] that specifically discussed inertial sensors and adaptive algorithms for gait analysis.

The differences between this review to the above works are this review aims to provide a comprehensive review of quantitative gait analysis using wearable sensors with no constraint on specific subject groups nor the method of analysis. Thus, it covers a wider range of gait analysis that includes various applications such in general walking gait, running gait, and different kinds of pathological gait. In addition to that, this review is intended to summarize the last decade's advancement of this topic, identify challenges in recent studies, and give future directions on research on this topic.

III. SYNTHESIS OF RESULTS

We ensured that all the included papers validated their proposed techniques by comparison against the gold standard



Fig. 3. Identified application of wearable-based gait analysis. Twenty-three studies discussed the application to both healthy control and patients with certain pathological conditions.

measurement such as motion capture and/or force plate systems or other widely accepted measurement methods, such as using an instrumented treadmill or the GAITRite® system. There were some exceptions to this criterion, such as studies that used force-based sensors [54], [94] on the foot to directly estimate the initial/terminal contact of the foot, analyzed only the statistical features from the raw data [55], employed a machine learning-based algorithm for classification [80], [111], or cited their previously validated framework [78], [88].

This section addresses several important points about wearable-based gait analysis, starting from the applications, sensors used and location attachment, gait features that can be extracted, data processing method, algorithms, as well as experiments and validation methods. Table I-IV summarize those important points from all of the eligible studies.

A. General Application of Wearable Gait Analysis

Based on the results of the literature search, we identified that 46 (61%) of the eligible studies tested their framework only on healthy control subjects, whereas 23 (30%) included both healthy control and patient groups. We did not discriminate the healthy subjects based on age; thus, children or the elderly were also considered as general applications. Analyses specific to the elderly [58], [62] or children [75], and comparisons between age groups such as normal middle-aged subjects and the elderly [49], [53], [61], [105] were examples of general application by age difference.

In terms of assessing different gait types, we identified a few studies that discussed both walking and running [38], [51], [91], [96]. The analysis of walking on different terrains such as ramp walkways, walking up- and downstairs, or outdoors was discussed in [68], [76], [96], [104], and [111]. Temporal, spatial, and spatiotemporal gait features were found to be the most commonly analyzed features within the studies. These features could be further derived into certain gait indices or gait metrics such as gait regularity and symmetry index [37], [102], which are mostly used in clinical applications. Estimation of lower limb joint angles [45], [51] and joint moments [50] were also conducted using wearable sensors.

B. Clinical Application of Wearable Gait Analysis

In addition to the general application of gait analysis, we identified several clinical applications such as

TABLE I
STATE OF THE ARTS OF QUANTITATIVE GAIT ASSESSMENT USING WEARABLE SENSORS

Ref.	Subjects (n)	Sensors (Location)	Experiments/ Benchmark	Algorithms/ Data Pro- cessing	Gait features [Performances metrics]
[36]	Healthy (3)	7 IMUs (lumbar, thighs, shanks, feet)	Overground indoor walk, variable-speed treadmill/ Motion capture	Non-ML/ Online (joint angles), Offline (stride length)	Stride length, joint angles (hip, knee, ankle) [RMSE < 4 deg]
[37]	Healthy (5)	1 IMU (left heel)	20 m corridor walk with various speed/ Camera	Non-ML (threshold based, sliding window)/ Real-time	Stride count, cadence, gait speed, ratio of swing/stance, stride regularity, stride length [err. $< 3\%$], walking distance [err. $< 2\%$]
[38]	Healthy (5)	2 IMUs (shanks)	Treadmill walk at [2,4] kph and run [8,12] kph/ Motion capture	Non-ML (threshold based)/ Offline	Gait events (HS and TO [err. $<1\%$], MSw) , stride time [err. $<1.6\%]$, swing, stance time.
[39]	Healthy (22)	1 acc. (L4-L5)	Level walk, 10m, barefoot/ Motion capture	Non-ML (peak detec- tion, integration)/ Of- fline	Step length, stride length $[p > 0.05]$, {stride, step, stance, swing, DS, SS} time, gait speed $[p > 0.05]$, cadence, foot symmetry
[40]	Healthy (10), OA(12)	4 IMUs (shanks and foot)	50 m corridor walk/ Motion capture	Non-ML/ Offline	Joint angles (multi-segment foot angles) [RMSE < 2 deg, Mean RoM diff. < 4 deg]
[41]	Healthy (10), pre-HD (10), HD (14)	l acc (thorax level)	4.8 m level walk/ GAITRite®	Non-ML (Inverted pen- dulum)/ Offline	Gait speed [ICC(CI) 0.95(0.75,0.97)], ca- dence [ICC(CI) 0.95(0.75,0.97)], stride length [ICC(CI) 0.89(0.77,0.95)], step length, step time, step time asymmetry, step-stride regular- ity
[42]	Healthy (8)	Textile socks	2 min walk on treadmill at 4 kph/ F-scan	Non-ML (direct use of sensors data)/ Offline	Gait events (HS [0.08 \pm 0.08 s], TO), stride time [err. $<$ 3.5%], stance time
[43]	Children (1), Children with CP (6)	7 IMUs (lumbar, thighs, shanks, feet)	10m walk preferred speed/ Motion capture	Non-ML ('Outwalk' protocol)/ Offline	Joint angles [RMSE < 4 deg]
[44]	Older (10), PD (10)	2 IMUs (upper shoes)	2x20 m and 4x50 m walk with 180 deg turn/ Motion capture	Non-ML/ Offline	Gait speed [2.8 \pm 2.4 cm/s], stride length [1.3 \pm 3.0 cm], swing width, path length
[45]	Healthy (5)	7 IMUs (pelvis, thighs, shanks, feet)	5m indoor level walk/ Mo- tion capture and force plates	Non-ML/ Offline	Joint angles [RMSE < 10.14 deg] and 2D joint trajectories
[46]	Healthy (10)	Instrumented in- soles	5m indoor level walk/ F- scan	Non-ML/ Real-time (Data streaming)	Step time, stance time, swing time
[47]	Healthy (10)	e-AR	Treadmill walk with vari- able speed and inclina- tion/ Instrumented tread- mill, High speed camera (250Hz)	Non-ML (recursive)/ Offline	Stride time [err. $< 1.47\%$], stance time [err. $< 4.84\%$], swing time [err. $< 8.03\%$]
[48]	Children (15), Children with CP (14)	2 IMUs (feet)	6 m straight walk and figure 8 walk for benchmark; 200 m walk in clinical setting with self selected speed for testing/ Motion capture	Non-ML/ Offline	Gait speed [4.3 \pm 4.2 cm/s], stride length [3.4 \pm 4.6 cm], % {stance, swing, DS}, cadence, heel-toe clearance, strike [0.5 \pm 2.9 deg] and lift off [3.9 \pm 5.8 deg] angles
[49]	Healthy (12), Older (12)	1 IMUs (L5)	5x25 m route on preferred and fast speed/ GAITRite®, camera	Non-ML (continuous wavelet transform)/ Offline	Step time and count, stride time [ICC 0.994-0.999], step length [ICC 0.756-0.929], step velocity [ICC 0.853-0.942]
[50]	Healthy (4)	3 IMUs (thigh [right], shank, foot) and force sensors	3 m walking straight indoor/ Motion capture and force plates	Non-ML/ Offline	Joint moments [3.5 % < NRMSE < 21 %]
[51]	Healthy (3)	Soft sensing suit	Instrumented split-belt treadmill with 5 speeds (3 walk and 2 run)/ Motion capture	Non-ML/ Offline	Joint angles (hip, knee, ankle) [RMSE < 15 deg]
[52] ^c	Healthy (10)	2 acc. (L3-L4 and shank) and 1 gyro (shank)	Instrumented split-belt treadmill with 5 speeds (all walk)/ Motion capture	Non-ML/ Offline	Gait events (IC, TC), stride time [RMSE $<$ 1.6%], step time [RMSE $<$ 4.3%], DS time [RMSE $<$ 25.7%], stance and swing time
[53]	Young (9), Mid-age (5), Old (6)	2 IMUs (ankles)	Self-selected speed walking with variation on stride ve- locity and stride length/ Mo- tion capture	Non-ML/ Offline	Heel [3.22 \pm 1.50 cm] and toe [1.69 \pm 0.70 cm] clearances, foot angle [2.49 \pm 1.21 deg]

TABLE II
STATE OF THE ARTS OF QUANTITATIVE GAIT ASSESSMENT USING WEARABLE SENSORS

Ref.	Subjects (n)	Sensors (Location)	Experiments/ Benchmark	Algorithms/ Data Pro- cessing	Gait features [Performances metrics]
[54] ^f	Healthy (10)	Sensorized insoles	Treadmill walk at 2 km/h/ Not needed	Non-ML/ Real-time	Gait phases
[55]	Healthy (25), SCD (25), PD (25)	2 IMUs (lower back, upper back)	6 min walking/ Not needed	Non-ML/ Offline	Statistical (Mean amplitude and Coefficient of Variation)
[56]	Older (19), PD (13)	3 IMUs (C7, L5, back of head)	2 min walking on 25 m cir- cuit/ GAITRite®	Non-ML/ Offline	Statistical (Magnitude, attenuation, harmonic ratio)
[57]	Healthy (15), PD (5), Stroke (4)	2 IMUs (feet)	10 m straight walk normal speed/ FSR sensors	Non-ML/ Online	Gait events (HS $[0.125 \pm 0.01 \text{ s}]$, TO $[0.089 \pm 0.015 \text{ s}]$), {stride, stance, swing, walking} time, stride count, stride length, gait speed, cadence
[58]	Older (82)	1 acc. (L3-L4)	2x40 m round walkway at comfortable walking speed/ GAITRite®	Non-ML/ Offline	(Cadence, gait speed, step length, step time) [ICC 0.91-0.96], variability and asymmetry in- dices
[59]	ACL (23), TKR (31)	e-AR	6 min walking on the corri- dor/ Pressure insoles	Non-ML/ Offline	Asymmetry(acc) [MSE=0.044]
[60]	Healthy (10), OA (14)	Smart shoes	15 m walking straight/ Force plates	ML (SVM)/ Real-time	Gait phases (IC, FF, HO, TO, swing) [Accuracy 94.08 %]
[61]	Healthy (10), Older (21)	1 acc. (waist)	10 m walking/ Video cam- era	ML (K-means cluster- ing)/ Offline	Gait event (IC), step detection [Sensitivity 99.33 %], frail classification
[62]	Older (24)	2 IMUs (shoes lateral)	5 min walking on treadmill with variable slope/ Instru- mented treadmill	Non-ML/ Offline	Gait speed [RMSE < 0.089 m/s], stride length [RMSE < 0.336 m], stride time [RMSE < 0.004 s], cadence [RMSE < 0.098 steps/min]
[63] ^f	PD (3)	Sensorized insoles, 6 IMUs (hips, shanks, feet)	18 m walking straight with variable pace/ Motion cap- ture	Non-ML/ Offline	Gait phases (IC, FF, TC, swing), step length [Mean diff. 0.9 cm]
[64]	Healthy (10)	Smart shoes	Free walking and 4x5 m walk/ Motion capture	Non-ML/ Real-time (Data streaming)	Gait phases (HS, stance, HO, swing) [Error range between 0.036-0.110 s]
[65]	Healthy (16), Neurologic patients (6)	2 IMUs (feet)	Walking with preferred pace/ Kinect	Non-ML/ Real-time (Data streaming)	Stride length [RMSE 0.05 m], step length, ca- dence, gait speed, {stride, stance, swing [RMSE 0.02 s]} time, foot clearance, turning rate
[66]	Healthy (12), Obese (10)	7 IMUs (pelvis, thighs, shanks, feet)	14 m walk straight/ STEP32	Non-ML/ Offline	Cadence, % gait phases, ROM (ankle, knee, hip) [Range of ICC 0.43-0.72]
[67]	Healthy (16), PD with FoG (26) non FoG (16)	2 IMUs (shins)	TUG (3 m walk)/ Motion capture and video camera (for FoG)	Non-ML/ Offline	Step velocity, stride length, stride time, ca- dence, FoG detection [Accuracy 98.51 %]
[68]	Healthy (6)	1 IMU (foot)	Level ground walking, stairs climbing/ Motion capture	Non-ML/ Offline	Step count, walking distance [Err. ;0.81 %], % gait phases, 3D trajectory [RMSE 0.28 m]
[69]	Healthy (7)	2 acc. (heels)	12 m walk/ Motion capture and video camera	Non-ML/ Offline	Gait events (HS [7.2 \pm 22.1 ms], TS [0.7 \pm 19.0 ms], HO [3.4 \pm 27.4 ms], TO [2.2 \pm 15.7 ms]), {stride, stance, swing} time [Range of ICC 0.87-0.98], heel clearance
[70]	Healthy (15), MS (45)	2 IMUs (shanks)	Various walking tasks/ Ac- tivity monitor, Inertial sen- sors	Non-ML/ Offline	Step count, {stride [6 \pm 9 ms], step [6 \pm 7 ms], swing [25 \pm 19 ms]} time
[71]	Healthy (25)	4 IMUs (ankles, mid of superior iliac spine, C2)	15 m walk normal speed/ Motion capture, GAITRite®	Non-ML/ Offline	Cadence [SEM 5.24 steps/min], gait speed [SEM 0.14 m/s], stride length [SEM 0.21 m], {stride [SEM 0.04 s], stance [SEM 0.04 s], swing [SEM 0.02 s]} time, {strike [SEM 2.11 deg], lift-off [SEM 3.33 deg], pelvis, spine} angle
[72]	Stroke (25)	1 IMU (L5)	Laboratory and real- life longitudinal study/ GAITRite®, OPAL, video camera	Non-ML/ Offline	Step velocity [ICC 0.744 m/s], step length [ICC -0.411 m], {step [ICC 0.797 s], swing [ICC 0.431 s], stance [ICC 0.759 s]} time, step width, asymmetry and variability indices

TABLE III
STATE OF THE ARTS OF QUANTITATIVE GAIT ASSESSMENT USING WEARABLE SENSORS

Ref.	Subjects (n)	Sensors (Location)	Experiments/ Benchmark	Algorithms/ Data Pro- cessing	Gait features [Performances metrics]
[73]	Healthy (14), children (10), CP (22)	3 IMUs (L2-L3, thighs)	30 strides at various speed/ FSR sensors	Non-ML/ Offline	Statistical (Pearson's r, variance ratio, harmonic ratio), Gait segmentation [Sensitivity 83.34 % (CP), 96.67 % (Healthy)]
[74]	Older (23)	2 IMUs (feet)	Combination of normal and fast speed with added cog- nitive task/ GAITRite®	Non-ML/ Offline	Cadence, gait speed [ICC 0.34-0.96], step time [ICC 0.22-0.27], step length [ICC 0.45-0.84]
[75]	Children (10)	6 IMUs (sternum, wrists, L4-L5, shanks frontal)	7 m walk straight at self- selected and fast speeds/ Motion capture	Non-ML/ Offline	Stride length [RMSE 6.43 % of subject's height], gait speed [RMSE 7.80 % of subject's height], {stride [RMSE 0.014 s], stance [RMSE 0.026 s]} time
[76] ^c	Healthy (20)	4 acc. (left wrist, waist, ankles)	Indoor, Outdoor, Treadmill walk and run/ FSRs sensors	Non-ML/ Offline	Gait events: (HS,TO) [F1 Scores (0.98,0.94) indoor, (0.82,0.53) outdoor]
[77] ^c	Healthy (35)	5 IMUs (L5, shanks, dorsal shoes)	2 min walk at 10 m path back and forth/ Force plates	Non-ML/ Offline	Gait events (HS, TO), {step, stance} time
[78]	AD (16)	1 acc. (L5)	Lab-based and free-living/ Prev. paper [5]	Non-ML/ Offline	Step velocity, step length, {step, swing, stance} time, step width, asymmetry and variability indices
[79]	Knee arthoplasty patients (16)	2 IMUs (below knees)	6 m walk straight at self- selected/ Motion capture	Non-ML/ Offline	Gait events (IC, TO), {stride [RMSE 0.036 s], stance [RMSE 0.041 s], swing [RMSE 0.049 s]} time
[80] ^c	Healthy (27), PD (27)	8 IMUs (chest, lumbar, thighs, shanks, feet)	15 m walk straight at self- selected and fast speeds/ Not needed	ML (SVM)/ Offline	Step length, gait speed, {step, stride} time, {hip, knee, ankle} ROM, Patient vs. control classification [Highest accuracy 79.96 %]
[81]	Healthy (30), Stroke (20), Joint disease (20)	7 IMUs (waist, thighs, shanks, feet)	Walk straight >15m on cor- ridor/ Motion capture	Non-ML/ Offline	Stride length, gait speed, stride freq., {stride, stance, swing} time, foot clearance, knee ROM [Position err. <0.015 m]
[82]	Healthy (24)	7 IMUs (waist, thighs, shanks, feet)	6 min walk test/ Motion capture	Non-ML/ Real-time	Step [RMSE 0.04 m] and stride length, step and swing width [RMSE 0.03 m], cadence [RMSE 3.1 steps/min], {step, stride, stance, SS, DS, swing} time [RMSE 0.02 s], gait speed [RMSE 0.03 m/s]
[83]	Healthy (10)	1 IMU (CoM)	Walking at self-selected speed/ Motion capture	Non-ML/ Offline	Step length [Abs. Err. 5.6 %], gait speed [Abs. Err. 13.5 %], walking time [Abs. Err. 14.9 %], walking distance
[84]	MS (4), HSP (9)	IMUs, FSRs	10 m walk, may use walking aid/ GAITRite®	Non-ML/ Offline	Gait events (IC, TC), % DS [3.89 \pm 2.61 %], DS time [0.064 \pm 0.060 s]
[85]	Healthy (16)	Flexible sensors	6 m walk at preferred speed/ Motion capture	Non-ML/ Real-time (Data streaming)	Knee angle [RMSE 1.2 \pm 0.4 deg]
[86] ^c	Healthy (15)	1 IMU (foot)	50 strides walk/ Motion capture	Non-ML/ Real-time (Data streaming)	Walking distance [Accuracy 95.24 %], stride count [Accuracy 95.47 %], {stride, stance, swing} time
[87]	Older (20)	1 acc. (waist)	Real-world setting/ Video camera	Non-ML/ Offline	Step detection, gait speed [Mean diff0.206 m/s (for speed < 1 m/s), and -0.045 (for speed 1-1.5 m/s]
[88]	Healthy (20), Neurological (20)	2 IMUs (ankles)	10 m corridor walk at pre- ferred speed/ Prev. paper	Non-ML/ Offline	Cadence, gait speed, stride length, stride time, $\%$ stance and swing, ankle ROM, gait symmetry and regularity [Position Error < 1%]
[89]	Healthy (20), Stroke (20), Joint disease (20)	6 IMUs (thighs, shanks, feet)	15 m obstacle-free corridor walk/ Motion capture	Non-ML/ Real-time (Data streaming)	Cadence, gait speed, stride length, stride time, % stance and swing, foot clearance, knee ROM and dorsiflexion-plantar angles [Err. < 3 deg]
[90]	Healthy (5)	2 IMUs (feet)	11 m walk indoor/ Motion capture	Non-ML/ Offline	Gait events (HS, TO), gait speed [Rel. Err. 6.3 ± 2.2 %], stride length [Rel. Err. 5.9 ± 3.3 %], % DS phase [Rel. Err. 4.3 ± 3.3 %]
[91]	Healthy (49)	2 IMUs (feet)	Treadmill run at comfort- able speed/ High speed camera	Non-ML/ Real-time	Step length [ICC 0.968-0.975], step frequency, stance [ICC 0.813-0.896] and swing [ICC 0.807-0.857] time
[92] ^f	Healthy (10)	4 IMUs (shanks, feet)	Treadmill walk with vari- able speeds/ Motion capture	Non-ML/ Real-time	Gait events (HS, TO), Gait phases (stance [Ac- curacy 97.9 %], swing [Accuracy 96.3 %]), ankle angle [RMSE 3.24 ± 0.67 deg]

TABLE IV
STATE OF THE ARTS OF QUANTITATIVE GAIT ASSESSMENT USING WEARABLE SENSORS

Ref.	Subjects (n)	Sensors (Location)	Experiments/ Benchmark	Algorithms/ Data Pro- cessing	Gait features [Performances metrics]
[93]	Stroke (25)	2 IMUs (feet)	Walking with total distance of 120 m/ Motion capture	Non-ML/ Offline	Stride time [Err. 0.003(0.020) s], stride length, cadence [Err0.341(0.467) steps/min], gait speed [Err. 0.002(0.003) m/s], % gait phases ; <i>Note</i> =[Err.=Non-paretic(Paretic)]
[94] ^f	Healthy (17), hemiparetic (18)	Textile insoles	20 m corridor walk back and forth at self-selected/ Not needed	Non-ML/ Real-time (Data streaming)	Plantar pressure, stride time, stride count
[95]	Healthy (30)	1 IMU (L5)	6 m indoor walk at preferred speed/ GAITRite®	Non-ML/ Offline	Gait speed [ICC 0.92], cadence [ICC 0.96], stride length [ICC 0.88], stride time [ICC 0.93], % gait phases (SS, DS, Swing, Stance) [ICC 0.18, 0.12, 0.47, 0.47]
[96]	Healthy (3)	3 IMUs (waist, thigh, shank)	Walk and run on treadmill, up and down stairs walking/ Footswitch insoles	ML (Random Forest)/ Offline	Gait phases (SS, DS, swing) [Accuracy 98.94 % (walk), 98.45 % (run)]
[97]	Healthy (22), cerebellar ataxia (29)	3 IMUs (chest, ankles)	5 m walk back and forth/ Motion capture	ML (Random Forest)/ Offline	Gait speed, cadence, gait ataxia quantification. [RMSE 0.18]
[98]	Healthy (10)	3 IMUs (chest, wrist, thigh), 4 FSRs	7 min treadmill walk at various speed/ Instrumented treadmill	Non-ML/ Real-time (Data streaming)	Gait events (HS, TO), stride time [RMSE 5.027 ms], stride count [Accuracy 99.6 %]
[99]	Healthy (9), PD (6)	2 IMUs (feet)	Walk at self-preferred speed/ GAITRite®	Non-ML/ Offline	Step length [Mean Err. 4.50 \pm 2.54 %] , step time [Mean Err. 2.97 \pm 2.51 %]
[100]	Healthy (5)	4 Ultrasonics sensors	8.5 m walking back and forth at normal speed/ Video camera	Non-ML/ Offline	Gait events, ankle angle [Mean diff. 0.19 \pm 1.19 deg], toe clearance [Mean diff. 0.02 \pm 0.84 cm]
[101]	Healthy (30), stroke (30)	8 IMUs (waist [left and right], knees, ankles, feet)	More than 15 m level ground corridor walk/ Mo- tion capture	Non-ML/ Offline	Stride length, cadence, gait speed, ankle ROM, gait symmetry, stance/swing ratio, foot clear- ance [Position Err. 0.02 m]
[102]	Healthy (6)	2 IMUs (shoes posterior)	4.5 m and 11 m indoor walk back and forth at self-preferred speed/ Mo- tion capture, force plates	Non-ML/ Offline	Gait events (IC [4.22 ± 15.48 ms], TO [-8.31 ± 21.02 ms], % gait phases, gait speed, stride length [Accuracy 93.23 %], heel clearance [2.22 ± 5.28 cm], {stride, stance, swing} time, variability and asymmetry indices
[103]	Healthy (3)	E-textile socks	11 m indoor walk at various speeds/ Motion capture	Non-ML/ Real-time (Data streaming)	% Gait phases, cadence, stride length [Pearson's $r = 0.283$], gait speed
[104]	Healthy (11)	7 IMUs (CoM, thighs, shanks, feet)	Walking on various terrain (level ground, stairs, ramp)/ Inertial sensors	ML (Neural Network)/ Real-time	Joint angles [NRMSE < 0.092 (knee joint)]
[105]	Healthy (20), older (20)	5 IMUs (chest, wrists, ankles)	Figure eight walk at normal speed on laboratory envi- ronment/ Motion capture	Non-ML/ Offline	Stide length [Abs. Err. 0.02 ± 0.03 m], gait speed [Abs. Err. -0.01 ± 0.02 m/s], step width, % gait phases, cadence, stride time, foot clearance, arm-related metrics
[106]	Healthy (5)	2 IMUs (top of shoes)	60 m total walking distance with variable stride length/ Optogait system	Non-ML/ Offline	Stride count, stride length [RMSE 5.0 cm], stride time [RMSE 0.04 s]
[107]	Healthy (8)	6 IMUs (thighs, shanks, feet)	4.5 m straight walk with variable speeds and tasks/ Motion capture	ML (kNN)/ Real-time	Stride time, stride length [RMSE 3.33 cm]
[108]	Healthy (9)	2 IMUs (lateral shoes)	20 m straight walk at pre- ferred speed/ Motion cap- ture	ML (Neural Network)/ Offline	Gait phases classification [Accuracy 92.63 %]
[109]	Healthy (40), PD (24)	1 IMU (L5)	10 m indoor straight walk at preferred speed/ GAITRite®	Non-ML/ Offline	Cadence [MSE Healthy(PD) 4.78(15.28) steps/min], gait speed [MSE Healthy(PD) 0.02(<0.01) m/s], stride time, stride length, % gait phases [MSE 17.40 % (for double support on PD)
[110]	Healthy (40)	Pressure insoles	10 m indoor walk with var- ious speeds/ GAITRite®	Non-ML/ Offline	Cadence [ICC 0.99], {stride, step, swing, stance, SS [ICC 0.65-0.96], DS [ICC 0.55-0.79]} time
[111]	Healthy (20)	1 IMU (ankle)	Walking on various terrain/ Not needed	ML (Random Forest)/ Offline	Gait activity classification [Accuracy 98.2 %]

Parkinson's Disease (PD), Huntington's Disease (HD), Cerebral Palsy (CP), Multiple Sclerosis (MS), Osteoarthritis (OA), post-stroke patients among much more clinical application within the literature. We found that most of the studies discussed the comparison from one disease group to the healthy control group or other disease groups. Other investigations on each of the mentioned disease groups are discussed below.

1) Parkinson's Disease: PD is a brain disorder that causes difficulties in walking, balance, coordination, and talking. Gait analysis performed in PD patients can provide several insights, such as the difference between the OFF and ON states of medication [44] and detection of freezing of gait (FoG) [67]. Quantitative gait analysis can provide clear quantifiable features that can track the progress of patients.

We found nine studies [44], [55]–[57], [63], [67], [80], [99], [109] that implemented a wearable sensor approach for gait analysis in PD patients. All of these studies used IMU sensors with a combination of two to eight IMU units. One study [63] proposed a combination of sensorized insoles and six IMUs to estimate the gait phases and step length while also providing rhythmic auditory feedback to the user.

The PD group has also been used as a disease control group to identify other neurological conditions such as spinocerebellar degeneration [55]. Other applications such as analysis of the upper body and postural control were discussed in [56], whereas the comparison of several machine learning-based classifications of PD patients was extensively discussed in [80].

2) Stroke: Stroke is caused by an interruption of blood supply to the brain, usually in the form of a blood clot. In the case of stroke or post-stroke patients, gait analysis is usually performed to track the rehabilitation progress. Six studies discussed the application of wearable gait analysis for stroke patients [57], [72], [81], [89], [93], [101]. A study used six IMUs on the lower limb [89] to provide a solution for tracking the rehabilitation progress of different groups of patients, including stroke patients, by estimating their pre- and post-treatment knee ROM.

All studies used various combinations of one to eight IMUs. A study, which used two IMUs on the feet [57], showed significant differences in several spatiotemporal features between stroke patients and healthy control groups. Another study [81], [101] also reported clear differences in gait features between the compared subject groups.

An extensive reliability and validation study of a single trunk-attached (L5) IMU [72] showed moderate to good agreement with the GAITRite® system in estimating the stance time and several step-based features. Another validation study using a foot-worn IMU [93] showed good to excellent agreement on various spatiotemporal gait features for both the paretic and non-paretic sides, and moderate agreement for the stance and swing phases.

3) Huntington's Disease: HD is a neurodegenerative condition characterized by progressive movement disorders. In [41], an accelerometer was used at the thorax level to extract various spatiotemporal features based on an inverted pendulum model [32]. A comparison between three distinct groups of healthy controls, pre-manifest HD, and manifest HD was presented. The results showed that there was a strong agreement between the sensor and the GAITRite® system, and the gait features extracted from the accelerometer proved effective in differentiating between the groups, especially the pre-manifest and manifest HD groups.

4) Cerebral Palsy: CP is the most common motor disability among children. CP is classified into three types: ataxia (poor balance and coordination), dyskinesia (uncontrollable movement), and spasticity (stiff muscles). Three studies [43], [48], [73] discussed the application of wearable sensors for gait analysis of CP patients.

In [43], a specific protocol was developed for children with CP, called the "Outwalk" protocol, to estimate the joint angles using seven IMUs. In [48], the authors pointed out significant differences between children with CP and typically developing children, as indicated by various spatiotemporal and kinematic features. Another study [73] showed that the three subclasses of CP subjects could be distinguished based on their motor function by analyzing their unique statistical features.

5) Multiple Sclerosis: MS is a disease that affects the central nervous system (brain and spinal cord) and causes communication problems between the brain and other body functions. Two studies [70], [84] employed wearable sensors for gait analysis in MS, as discussed below. Both studies employed IMU sensors; in [70], two units were used on the shanks, and in [84], an IMU was combined with three force sensing resistor (FSR) sensors inside the insole to form a wearable system.

In [70], various walking protocols were designed with variable walking speeds to assess the MS group. The gait event detection algorithm was adapted from [33] and the gait parameters were further categorized into temporal features. Their proposed framework was validated against an activity monitoring system (GT3X) and a commercially available inertial sensor (MTx). The results showed that there were $2 \pm 2, 6$ \pm 9 ms, and 25 \pm 19 ms errors in the step count, stride time, and swing time, respectively. By comparing the MS group with the healthy control group, it was found that the proposed system could detect significant (p < 0.01) and distinct gait characteristics between the two groups. The authors of [84] proposed the use of sensorized insoles consisting of three FSR sensors and an IMU to extract the temporal gait features of patients with gait disorders such as MS and hereditary spastic paraplegia. Initial contact (IC) and terminal contact (TC) events were labeled manually based on the heel FSR and lateral FSR, respectively. The results showed that there was a mean error of 64 ± 60 ms in the double support (DS) time, expressed as $3.89 \pm 2.61\%$ of the DS phase by analyzing 1321 strides of all participants.

From both studies, we found that temporal features were the main gait features used to characterize MS patients. Several temporal features, such as stride time and step time, were found to be effective in differentiating between healthy and MS subject groups. Moreover, it was demonstrated that temporal features could also distinguish MS subjects based on the disease severity level [70].

6) Joint Diseases: Articles on various joint-related diseases such as osteoarthritis (OA) were found in the search pool.

OA is a degenerative joint disease, which occurs when the protective cartilage at the ends of bones is worn down. In this case, gait analysis was performed to assess the effectiveness of the treatment plan. Six studies have discussed various joint-related diseases, such as OA [40], [60], anterior cruciate ligament (ACL) reconstruction [59], knee arthroplasty [79], and other joint diseases [81], [89].

The estimation of kinematic features such as foot angles was found to be effective in differentiating between healthy control and ankle OA groups [40]. A study using an ear-worn accelerometer [59] provided various asymmetry metrics to monitor the recovery gait of patients who underwent ACL reconstruction or total knee replacement. Integrated FSRs and IMU sensors on smart shoes were used to extract the gait phases of OA patients by using the SVM algorithm in real time with 94 % accuracy [60]. Two IMUs placed below the knees were used to extract the temporal features in [79] for the analysis of knee arthroplasty patients after knee replacement surgery. The studies [81] and [89] used seven and six IMUs, respectively, and extracted the gait features, which mainly comprised the temporal, spatial, spatiotemporal, and kinematic features. Knee ROM was chosen as one of the primary features to track the prior and post-treatment gaits of the patients.

C. Sensors and Their Locations

In this review, we simplify the sensor categorization to IMU-based, non-IMU, and a combination of IMU and other sensors (IMU+). IMU-based sensors were found to be the most common wearable sensors used in this study. An IMU generally consists of an accelerometer, a gyroscope, and a magnetometer. Using all three components can provide the orientation and relative position information by means of the sensor fusion method. Nevertheless, in this review, we found that these three components are not always used together. Some studies used only the accelerometer [39], [55], [56], whereas others used only the gyroscope unit [38]. In addition, we found unique sensor systems such as textile-based [42], [103], ultrasonic-based [103], and flexible [85] sensors.

In terms of sensor location, we found that 42, 17, and 12 studies proposed the attachment of the sensor on the lower limb, both the trunk and lower limb, and only the trunk, respectively. From the 42 studies that proposed lower-limb attachment, we found that 31 of those studies proposed the foot-based sensors which accounts for either ankle, instep, heel-level, or lateral foot sensor attachment, or could also be in the compact form of either socks, insoles, or smart shoes. Only two papers proposed a head-attached sensor, both of which were from the same authors who proposed an ear-attached sensor [47], [59]. Interestingly, two papers proposed the attachment of the sensors on the upper limb, trunk, and lower limb; reference [75] focused on the gait assessment in children, whereas [76] compared several algorithms and tested them on various terrains/environments. One study proposed a combination of sensors attached on the trunk and head [56] for the assessment of gait in elderly and PD patients. The distribution of the sensors and their locations are depicted in Fig. 4.



Fig. 4. (a) Distribution of wearable sensors used, where IMU-based sensors account for 80%, and IMU+ which is combination of IMU and other sensors account for 5% of the total eligible studies. (b) Location of sensor attachment to the body, where the most was found at lower limbs (42 studies) followed by lower limbs and trunk attachment (17 studies). Overall, foot-based sensor attachment (31 studies) was favored over the last ten years. (c) The detailed view of attachment of sensors in different region of human body as summarized from the eligible papers.



Fig. 5. Classification of gait features that are extracted from wearable sensors, which comprised of gait event/gait phases, spatial, temporal, spatiotemporal, gait index/metrics, biomechanical features, statistical features. In this review, biomechanical features are any kinematic or kinetic-based gait features such as joint angles and joint moment.

D. Gait Features

In this review, gait features are quantifiable parameters or characteristics of gait that are measured directly from wearable sensors or estimated through sets of algorithms. We classified the gait features into seven categories, as depicted in Fig. 5. The first category is the gait event or gait phase. Gait events are instantaneous events that occur based on the set detection algorithm, whereas the gait phase marks the time period between the initiation and termination of that phase. The most important gait events are IC and TC, which represent the instants when the stance phase and swing phase are about to commence, respectively. In some of the identified studies, IC and TC may also be called heel-strike (HS) and toe-off (TO), respectively. Other important gait events identified throughout the review are midswing (MSw), toe-strike (TS), and heel off (HO). These events were used as markers to identify gait phases, which under the new terms of gait classification are classified into eight distinct phases: IC, loading response, mid-stance, terminal stance, pre-swing, initial swing, MSw, and terminal swing. In addition to those phases, we also identified single support (SS), double support (DS), and foot flat (FF), in the eligible studies.

The second category encompasses the temporal features, which contain time-based information. Stride time, step time, stance time, swing time, SS time, DS time, and cadence are the widely known temporal gait features. Usually, these features are computed using information from gait event detection. For example, stride time, which is defined as the time taken to complete a gait cycle, can be estimated from the results of the HS or TO event. The stance time is estimated from the point of IC or HS to the point of TC or TO. The SS and DS times are estimated from the information corresponding to both sides of the foot, where the SS time is equivalent to the time elapsed when only one foot is in contact with the ground. The DS time is the time elapsed when both feet are in contact with the ground. Cadence is the number of steps walked per minute. Other than these features, some studies also computed the total walking time, which is the sum of the stride times per walking trial.

The third category encompasses the spatial features that contain the length-based information, such as stride length and step length. In addition, we also identified several spatial features such as swing width, path length, walking distance, traveled arm distance, and foot clearance (heel and toe).

The fourth category encompasses the spatiotemporal features, which are derived based on both spatial and temporal features. In this case, the gait speed and step velocity were defined as the spatiotemporal gait features.

The fifth category encompasses the biomechanical features, which consist of kinematic and kinetic features. The joint angles such as the hip, knee, and ankle angles are the most commonly analyzed kinematic gait features. In addition, we also identified the strike and lift-off angles, which are the angles at which the foot is about to contact the ground and is lifted from the ground, respectively. We found a clinical gait analysis study, which extracted and used the pelvis and spine angles of all axes [71] for their analysis. In another study, we noticed the use of the turning rate feature to assess the turning movement performed by the subject. Regarding dynamic features, we identified some studies that estimated the joint moments and extracted the plantar pressures from pressure-based insoles.

The sixth category encompasses the statistical features derived from the raw sensor data. We observed that the magnitude, RMS value, harmonic ratio, mean amplitude, coefficient of variation, Pearson coefficient, and variance ratio were extracted and classified as the statistical features within the reviewed studies.

The seventh category encompasses the gait index or gait metrics, which are unique features derived from any of the major gait features mentioned previously. For example, the variability, symmetry, and asymmetry indices were identified based on the step length, stance time, and swing time. Another commonly reported gait metric is the stride or step count, which is the total number of strides or steps in a particular walking trial. Stride and step regularity, ratio of swing and stance, joint trajectory, and stride frequency were also used in the eligible studies. Regarding clinical gait metrics, we found a few studies that discussed frailty classification in the elderly [61] and the FoG metric in PD patients [67].

E. Data Processing

Raw data from sensors can be processed in an offline, online, or real-time manner. We found that most of the studies in this review adopted offline data processing. This may have been due to the need for proper experimental procedures designed carefully to validate the proposed framework against the gold standard system such as motion capture and force plates. Online data processing was presented in two studies [36], [57]. The first study [36] used seven IMUs attached to the lumbar, thighs, shanks, and feet to estimate the joint angles in an online manner. The integral differences in the angular velocity between the two sensor locations, coupled with the Kalman filter (KF), was proposed to estimate the joint angles. The results showed that the root mean squared error (RMSE) of the joint angle estimation varied depending on the walking speed, where the slowest walking speed resulted in the lowest RMSE. The second study [57] used a threshold-based algorithm to estimate the HS and TO events from a foot-mounted IMU. Several spatiotemporal features were derived and compared between the subject groups by conducting a 10 m walking experiment followed by online data processing. Their method was validated against the results of FSR sensors and by comparison with the results of some existing studies. The proposed method showed good agreement with the FSR sensor-based method and was comparable to the existing method. Moreover, some of the extracted spatiotemporal features showed a significant difference (p < 0.05) between the stroke patients and healthy controls, and PD patients and healthy controls.

Real-time data streaming has become more common in recent years [46], [64], [65], [85], [86], [89], [94], [98], [103] and has contributed to improved processing time efficiency. Real-time data processing was implemented in [37], [54], [60], [82], [91], [92], [104], [107]; in [104], up to seven IMU units were used for lower limb joint angle estimation. We briefly discuss the studies that have implemented real-time data processing. In [37], one IMU was attached to the left heel to extract the spatiotemporal gait features such as cadence, velocity, stride length, and walking distance. Threshold-based algorithms and sliding window techniques enabled real-time implementation. Real-time data processing was employed to estimate the gait phases from an insole equipped with 64 pressure-sensitive elements, which simultaneously recorded the GRF and center of pressure [54].

Smart shoes equipped with seven FSRs and one accelerometer on each shoe were proposed to estimate the gait phases in real time using a machine-learning-based algorithm [60]. An investigation of the validity and test-retest reliability of real-time event detection and further spatiotemporal gait feature estimation using seven IMUs were presented in [82]. The validity of two commercial wearables, Stryd and RunScribe, for performing gait analysis was presented in [91], which were benchmarked against a high-speed video analysis recorded at 1000 Hz. The use of four IMUs attached to the shank and foot of both legs for gait phase recognition and ankle angle estimation in real time was studied in [92]. Another study [104] used seven IMUs placed near the center of mass, both thighs, shanks, and feet to estimate the joint angles in real time. Finally, an open-source application on Python for gait analysis using six IMUs attached to the thighs, shanks, and feet was proposed in [107]. kNN followed by correction using foot acceleration was used as a gait segmentation algorithm to estimate the TO events. A GUI was developed for the commands and visualization tools.

F. Algorithms

Treatment of raw data, or data pre-processing, usually involves applying filtering techniques to remove unnecessary data or noise from the raw data. For example, Butterworth low-pass filters with various orders and cut-off frequencies were commonly applied in these studies. The sliding window technique was implemented in studies proposing a real-time processing approach [37] or in data preparation for threshold-based algorithms [76], [108].

We identified various sets of algorithms for processing the raw data from wearable sensors to perform quantitative gait analysis. To simplify, we classified the main algorithm into two classes: ML-based algorithm and non-ML-based (conventional) algorithm. Most eligible studies employed the conventional algorithm to extract the gait features. Regarding gait event/gait phase detection, we identified several algorithms such as peak detection algorithm, threshold-based algorithms, state machines, heuristic rule-based algorithm, and finding of local minima/local maxima, which have nearly the same principles. A fast Fourier transform (was implemented to extract the gait frequency and cadence in a few studies [57], [97]. Other methods such as continuous wavelet transform have been implemented to extract the gait events [49], [72], [78] and perform stride segmentation [69].

Because the majority of the eligible papers used IMU-based sensors, several sensor fusion algorithms were applied to estimate the orientation of the sensor position. The KF and complementary filter are the two most adopted orientation estimation methods. This orientation position combined with the knowledge of gait events or gait phases can provide spatial features such as the stride length and step length, listed in section III-D. Double-integration combined with the zero-velocity update method was implemented in several studies to estimate the stride length. A model-based approach such as an inverted pendulum [41], [49], [72], [78], [83], [103] was also used in a few studies to estimate the gait features such as stride length.

Several studies have implemented ML-based algorithms such as SVM [60], [80], [104], RF [96], [97], [111], NN [96], [104], [108], kNN [80], [107], k-means clustering [61], decision tree [80], [104], logistic regression [96], naïve Bayes, and LDA [80] in their proposed framework for quantitative gait analysis using wearable sensors. Most of the studies used ML-based algorithms for classification problems such as gait event/gait phase classification [60], [80], [96], [108], and classification of gait on different terrains [111]. Further, we identified four studies that employed an ML-based algorithm as a supporting algorithm for performing operations such as crosschecking [61], correction [104], and false phase detection [107]. Lastly, only one study has proposed a scoring function for gait ataxia [97].

G. Experiments and Validation

Experiments to capture the gait can be performed in various ways. The experimental protocols were mostly designed to suit the objective of gait analysis. Therefore, no experimental protocol can be deemed better than the rest. In this review, we present various experimental protocols developed in the identified studies, as summarized in Table I-IV. Indoor-level ground walking was the most commonly adopted protocol. Other experiments such as running on a treadmill operated at various speeds, outdoor walking, figure-eight walk, timed up and go, walking on a ramp, and walking on stairs with different distances covered or different number of repetitions were also identified. We found that the design of the experimental protocols was highly dependent on the environmental constraints, subjects in the study, and types of wearable sensors.

We noticed several validation systems that are considered the gold standard for assessing gait, such as the motion capture system, GAITRite®, instrumented treadmill, and force plates. Besides these systems, some studies have validated their proposed wearable gait analysis framework against camera-based systems, pressure-sensor systems, and other inertial sensor-based systems. Each of the studies has its own method of reporting the benchmark or validation results. The summary of performance metrics from each of the studies are presented in Table I-IV. We identified several validation metrics such as the absolute error, relative error, accuracy and precision, RMSE, mean differences, student's t-test (p-values), intra-class correlation (ICC), confidence interval, limit of agreement, Bland-Altman plot, sensitivity and specificity, F1 score, false-positive rate, and standard error of mean. The broad range of validation metrics has made reporting a meta-analysis of the benchmark studies impractical.

IV. DISCUSSION

A. Inferences Drawn From Wearable-Based Gait Analysis for General Application

As discussed in Section III, 91 % of the eligible papers implemented their proposed approach on a healthy group or demonstrated a general application. We found that the healthy group mostly served as the control variable for comparison with other pathological groups. Temporal, spatial, and spatiotemporal gait features were found to be the most common assessments for general application. These features were not only reported as obtained, but could also be utilized to determine gait index features such as gait regularity or gait symmetry [88]. A few papers have discussed the difference between age groups, that is, healthy adult subjects and elderly subjects [49], [61], [105]. Regarding sports applications, running-based studies were discussed in [38], [51], [91], and [96]. Gyroscope-based temporal gait analysis with a peak running speed of 12 km/h was validated in [38]. Commercial wearables such as RunScribe and Stryd were evaluated to be effective wearables for measuring the spatiotemporal gait features while running at a comfortable speed [91]. In another study, the spatiotemporal features such as gait speed were associated with survival in older adults [115]. The importance of foot clearance assessment among elderly subjects was explained in [112].

B. Most Influential Features and Challenges in Wearable-Based Gait Analysis for Clinical Application

No particular gait feature can be considered superior to the rest for explaining certain gait conditions of a subject.

Each gait condition has its own unique features or markers that are distinguishable from the other conditions. In certain cases, a visual inspection can be sufficient to assess certain pathological markers such as hemiparesis, where one side of the body is weakened. However, a quantifiable parameter is required to report the factual condition of the subject. Temporal gait features such as stride time or step time were found to be effective in differentiating between three types of MS patients based on the severity level of the disease [70]. Detection of FoG, as discussed in [67], was found to be helpful in the assessment of PD patients. Statistical features, as described in [73], were found to be successful in detecting gait impairment in CP subjects. Other studies have shown that the knee ROM feature can be used to track the rehabilitation progress of arthropathy and stroke patients [89]. Furthermore, combining gait and other motions may provide a more comprehensive assessment of the motor performance [117].

We identified various clinical applications such as PD, MS, and post-stroke pathology, as depicted in Fig. 3. The performance, validity, and reliability of each proposed wearable approach have been extensively investigated. Because each clinical condition may produce a unique gait pattern disorder, we presume that it will affect the performance of any of the wearable-based approaches considered in this review. This issue highlights the importance of benchmark experiments before applying the proposed approach to new applications.

C. Challenges in Wearable Sensors Application and the Importance of Sensor Placement

We found that gyro drift is a major concern when using IMU-based sensors. There are several methods for reducing or removing the drift, such as using the KF [36], [65], [68], [104] and applying zero velocity update [65], [81], [90], [101], which are widely implemented in the studies examined in this review. Environmental interference, which affects the magnetometer, is also a problems observed in this review. A simple calibration procedure, which is easy to implement, was proposed in [81] to compensate the interference. Concerns relating to the secure attachment of the device to the body have also been raised in several studies. Soft-tissue artifacts have been found to be a source of noise in data processing [43], [66]. Some studies that used multiple IMUs were required to perform sensor-to-segment alignment and calibration [45], [92] to achieve better performance. This may increase the preparation time but is comparatively faster than using marker-based systems.

Regarding placement of the sensor, most studies placed the sensors on the lower limbs, followed by the lower trunk in the lumbar vertebrae region. To emphasize more on the lower-limb region attachment, we observed a trend of foot-based sensor attachment for wearable-based gait analysis over the past ten years, which account for 31 studies in this review. These positions were found to be well suited for wearable-based gait analysis in terms of the number of features that can be extracted as well as the accuracy of gait feature estimation, where attaching the sensor to the foot yielded better results of

26483

gait event detection than attaching to the shank or trunk [77]. The number and location of sensors were found to be more influential factors than the algorithm used for discriminating the severity stages of PD patients [80]. The search for the optimal location of foot-based sensors was investigated in [86], where the medial aspect of the foot, followed by the posterior side of the foot, were identified as the optimal locations for stride-related feature extraction.

D. Contribution of Machine Learning Algorithms in Wearable Sensor Gait Analysis and Current Challenges

Machine learning algorithms have been employed in several studies, where they have been mostly used for classification problems such as gait events or gait phase classification [60], [80], [96], [108] and gait activity/terrain classification [111]. We also identified a few other applications such as verification of frailty discrimination [61], joint angle correction [104], false swing phase detection [107], and scoring algorithms in gait ataxia [97]. With our search strategy, we did not find any paper on machine-learning algorithms for extracting gait features. However, we found a recent study [116] outside our search criteria, which implemented a deep learning approach to estimate the spatiotemporal gait features. The dataset for deep learning and the ground truth were collected from seven IMUs and motion capture combined with force plate systems, respectively, with multiple 5 m walk protocols at various speeds. Another challenge related to machine learning is the availability of datasets for various sensor positions and the validated ground truth.

E. Online and Real-Time Data Processing for Wearable-Based Quantitative Gait Analysis

We found several studies that implemented online or real-time data processing, as discussed in Section III-E. The vast development in computing power has enabled easier implementation of online and real-time data processing. This provides several advantages for both general users of wearable devices and for patients and clinicians in clinical settings. For example, in sports applications, real-time data processing enables the user to directly reflect and possibly correct their actions in real time for motion improvement based on the received real-time insights. In the clinical setting, real-time visualization is favored as real-time insight for both patients and clinicians. Real-time data processing also enables real-time feedback to the user, which may act as an intervention in the case of FoG in PD patients. Nevertheless, the implementation of online or real-time data processing may face problems such as data package loss while transmitting the data to the processing unit; therefore, such methods must be validated first to ensure the accuracy of the extracted gait features.

F. Slow Walking Speed and Its Effect to the Performance of Algorithms

Slow walking speed is often observed in elderly subject and pathological groups. In some of the studies, we observed that the stride detection performance was decreased at a slow walking speed. For example, a study reported a 97.8 % mean accuracy on 1.0 km/h speed, while 99.9 % detection accuracy was observed from speed 1.5 - 4.0 km/h [98]. Another example relates to the ICC metric, where slow walking speed produced the lowest ICC value when compared with the comfortable and maximum walking speeds [110]. The algorithm design and evaluation process should be improved to accommodate slower walking speeds for gait analysis in the elderly group.

G. Issues Related to Experiment Protocols and Validation Metrics

Each of the reviewed studies proposed various experimental protocols. Most of it can be simplified and referred to as "level ground walking," performed in a laboratory, long corridor, or outdoor setting. Pragmatically, a longer walkway is more suitable for capturing the natural gait of a subject, as it provides adequate time and distance for the subjects to adjust their walking as desired. This study [119] concluded that a minimum of 25 and 33 strides are required to properly compute the step symmetry and stride regularity, respectively, in healthy control subjects. Regarding the experimental protocol, a study [120] suggested that curved walking instead of straight walking is more appropriate for assessing people with gait disorders. However, a treadmill-based experiment may affect the natural gait of the subject. Further, the authors of [121] and [122] suggested that self-paced instead of fixed speed walking allows more natural stride variability.

We found that each of the studies adopted various validation metrics. This made it difficult to quantitatively compare the performances of the proposed frameworks. For example, the gait event detection was validated based on the absolute error, relative error, accuracy, and precision, as well as the ICC metric. Therefore, researchers working in this area may need to report their benchmark and validation studies in terms of various metrics to allow comparison with the existing stateof-the-art methods.

H. Integration of Wearable Sensor Gait Analysis and Feedback System

In this review, several studies were found to analyze the data from wearable sensors and provide them as feedback to the user through various modalities such as audio, visual, and haptic feedback [54], [63], [92], [94]. Feedback strategies are needed for gait retraining or real-time assessment to correct certain parameters in rehabilitation or sports applications. Notwithstanding the studies considered in this review, several studies have demonstrated immense progress made in the wearable-based gait feedback system. The auditory feedback investigated in [118] was found to be effective in the short-term rehabilitation of stroke patients with hemiparesis. A biofeedback system for gait and balance training in PD subjects was proposed in [17] and [18]. In another study [113], a sensing sock device with smartphone-based feedback based on various modalities, including combining several feedback strategies such as auditory, visual, and haptic feedback, was proposed. By comparing different feedback modalities, a study concluded that real-time haptic feedback was effective and less expensive than visual feedback [114] for gait retraining of runners with high tibial load. Nevertheless, the effectiveness and correctness of the feedback strategies need to be investigated further.

V. CONCLUSION

This review investigated the use of wearable sensors for quantitative gait analysis. The proposed frameworks for wearable gait analysis, identified in this review, must be validated against gold standard measurements or other established sensor systems to ensure the accuracy of the acquired measurements. The correlation between the number and types of sensors along with the location of attachment to the body, proposed method and algorithms used, and number and types of quantitative gait features that can be extracted were discussed comprehensively in this review. Future research should explore the integration between wearable-based gait assessment and feedback systems to provide real-time feedback to users and patients. Another area is to investigate the performance and validity of wearable-based gait analysis in other clinical applications. The use of machine learning algorithms to quantify the gait features and achieve certain performance scores for clinical application is also a challenge but necessary for researchers pursuing this research field.

REFERENCES

- Y.-S. Lee, C.-S. Ho, Y. Shih, S.-Y. Chang, F. J. Róbert, and T.-Y. Shiang, "Assessment of walking, running, and jumping movement features by using the inertial measurement unit," *Gait Posture*, vol. 41, no. 4, pp. 877–881, May 2015.
- [2] Q. Mei, J. Fernandez, W. Fu, N. Feng, and Y. Gu, "A comparative biomechanical analysis of habitually unshod and shod runners based on a foot morphological difference," *Hum. Movement Sci.*, vol. 42, pp. 38–53, Aug. 2015.
- [3] B. H. Dobkin, X. Xu, M. Batalin, S. Thomas, and W. Kaiser, "Reliability and validity of bilateral ankle accelerometer algorithms for activity recognition and walking speed after stroke," *Stroke*, vol. 42, no. 8, pp. 2246–2250, Aug. 2011.
- [4] C. Punin *et al.*, "A non-invasive medical device for Parkinson's patients with episodes of freezing of gait," *Sensors*, vol. 19, no. 3, p. 737, Feb. 2019.
- [5] S. D. Din, A. Godfrey, and L. Rochester, "Validation of an accelerometer to quantify a comprehensive battery of gait characteristics in healthy older adults and Parkinson's disease: Toward clinical and at home use," *IEEE J. Biomed. Health Informat.*, vol. 20, no. 3, pp. 838–847, May 2016.
- [6] G. Rescio, A. Leone, and P. Siciliano, "Supervised machine learning scheme for electromyography-based pre-fall detection system," *Exp. Syst. Appl.*, vol. 100, pp. 95–105, Jun. 2018.
- [7] T. Virmani, H. Gupta, J. Shah, and L. Larson-Prior, "Objective measures of gait and balance in healthy non-falling adults as a function of age," *Gait Posture*, vol. 65, pp. 100–105, Sep. 2018.
- [8] M. B. Nebel *et al.*, "The relationship of self-reported pain and functional impairment to gait mechanics in overweight and obese persons with knee osteoarthritis," *Arch. Phys. Med. Rehabil.*, vol. 90, no. 11, pp. 1874–1879, Nov. 2009.
- [9] M. Benedetti, V. Agostini, M. Knaflitz, V. Gasparroni, M. Boschi, and R. Piperno, "Self-reported gait unsteadiness in mildly impaired neurological patients: An objective assessment through statistical gait analysis," J. Neuroeng. Rehabil., vol. 9, no. 1, p. 64, 2012.
- [10] L. C. Benson, C. A. Clermont, E. Bošnjak, and R. Ferber, "The use of wearable devices for walking and running gait analysis outside of the lab: A systematic review," *Gait Posture*, vol. 63, pp. 124–138, Jun. 2018.
- [11] G. Quer *et al.*, "Wearable sensor data and self-reported symptoms for COVID-19 detection," *Nature Med.*, vol. 27, no. 1, pp. 73–77, Jan. 2021.

- [12] J. Dunn *et al.*, "Wearable sensors enable personalized predictions of clinical laboratory measurements," *Nature Med.*, vol. 27, no. 6, pp. 1105–1112, Jun. 2021.
- [13] D. Trojaniello, A. Ravaschio, J. M. Hausdorff, and A. Cereatti, "Comparative assessment of different methods for the estimation of gait temporal parameters using a single inertial sensor: Application to elderly, post-stroke, Parkinson's disease and Huntington's disease subjects," *Gait Posture*, vol. 42, no. 3, pp. 310–316, 2015.
- [14] S. A. A. N. Bolink *et al.*, "Assessment of physical function following total hip arthroplasty: Inertial sensor based gait analysis is supplementary to patient-reported outcome measures," *Clin. Biomech.*, vol. 32, pp. 171–179, Feb. 2016.
- [15] F. De Cillis, F. De Simio, and R. Setola, "Long-term gait pattern assessment using a tri-axial accelerometer," J. Med. Eng. Technol., vol. 41, no. 5, pp. 346–361, Jul. 2017.
- [16] A. Rodríguez-Molinero *et al.*, "Analysis of correlation between an accelerometer-based algorithm for detecting Parkinsonian gait and UPDRS subscales," *Frontiers Neurol.*, vol. 8, p. 431, Sep. 2017.
- [17] I. Carpinella *et al.*, "Wearable sensor-based biofeedback training for balance and gait in Parkinson disease: A pilot randomized controlled trial," *Arch. Phys. Med. Rehabil.*, vol. 98, no. 4, pp. 622–630, Apr. 2017.
- [18] B. M. Bartels, A. Moreno, M. J. Quezada, H. Sivertson, J. Abbas, and N. Krishnamurthi, "Real-time feedback derived from wearable sensors to improve gait in Parkinson's disease," *Technol. Innov.*, vol. 20, no. 1, pp. 37–46, Nov. 2018.
- [19] P. P. Panciani, K. Migliorati, A. Muratori, M. Gelmini, A. Padovani, and M. Fontanella, "Computerized gait analysis with inertial sensor in the management of idiopathic normal pressure hydrocephalus," *Eur. J. Phys. Rehabil. Med.*, vol. 54, no. 5, pp. 724–729, Sep. 2018.
- [20] L. Angelini et al., "Is a wearable sensor-based characterisation of gait robust enough to overcome differences between measurement protocols? A multi-centric pragmatic study in patients with multiple sclerosis," *Sensors*, vol. 20, no. 1, p. 79, Dec. 2019.
- [21] S. Shirai *et al.*, "The responsiveness of triaxial accelerometer measurement of gait ataxia is higher than that of the scale for the assessment and rating of ataxia in the early stages of spinocerebellar degeneration," *Cerebellum*, vol. 18, no. 4, pp. 721–730, Aug. 2019.
- [22] A. Liberati, "The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: Explanation and elaboration," *Ann. Int. Med.*, vol. 151, no. 4, pp. e1–e34, Aug. 2009.
- [23] D. Kobsar *et al.*, "Wearable inertial sensors for gait analysis in adults with osteoarthritis—A scoping review," *Sensors*, vol. 20, no. 24, p. 7143, Dec. 2020.
- [24] I. T. G. de Oliveira Gondim, C. D. C. B. de Souza, M. A. B. Rodrigues, I. M. Azevedo, M. D. G. W. de Sales Coriolano, and O. G. Lins, "Portable accelerometers for the evaluation of spatio-temporal gait parameters in people with Parkinson's disease: An integrative review," *Arch. Gerontol. Geriatrics*, vol. 90, Sep. 2020, Art. no. 104097.
- [25] P. R. F. Junior, R. C. F. D. Moura, C. S. Oliveira, and F. Politti, "Use of wearable inertial sensors for the assessment of spatiotemporal gait variables in children: A systematic review," *Motriz, Revista Educação Física*, vol. 26, no. 3, pp. 1–11, 2020.
- [26] S. Díaz, J. B. Stephenson, and M. A. Labrador, "Use of wearable sensor technology in gait, balance, and range of motion analysis," *Appl. Sci.*, vol. 10, no. 1, p. 234, Dec. 2019.
- [27] P. Dasgupta, J. VanSwearingen, A. Godfrey, M. Redfern, M. Montero-Odasso, and E. Sejdić, "Acceleration gait measures as proxies for motor skill of walking: A narrative review," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 29, pp. 249–261, 2021. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/9293160, doi: 10.1109/TNSRE.2020.3044260.
- [28] A. Saboor *et al.*, "Latest research trends in gait analysis using wearable sensors and machine learning: A systematic review," *IEEE Access*, vol. 8, pp. 167830–167864, 2020.
- [29] W. Tao, T. Liu, R. Zheng, and H. Feng, "Gait analysis using wearable sensors," *Sensors*, vol. 12, no. 12, pp. 2255–2283, 2012.
- [30] I. H. Lopez-Nava and A. Munoz-Melendez, "Wearable inertial sensors for human motion analysis: A review," *IEEE Sensors J.*, vol. 16, no. 22, pp. 7821–7834, Nov. 2016.
- [31] R. Caldas, M. Mundt, W. Potthast, F. B. de Lima Neto, and B. Markert, "A systematic review of gait analysis methods based on inertial sensors and adaptive algorithms," *Gait Posture*, vol. 57, pp. 204–210, Sep. 2017.

26485

- [32] W. Zijlstra and A. L. Hof, "Assessment of spatio-temporal gait parameters from trunk accelerations during human walking," *Gait Posture*, vol. 18, no. 2, pp. 1–10, Oct. 2003.
- [33] K. Aminian, B. Najafi, C. Büla, P. F. Leyvraz, and P. Robert, "Spatiotemporal parameters of gait measured by an ambulatory system using miniature gyroscopes," *J. Biomech.*, vol. 35, no. 5, pp. 689–699, 2002.
- [34] S. O. H. Madgwick, A. J. L. Harrison, and R. Vaidyanathan, "Estimation of IMU and MARG orientation using a gradient descent algorithm," in *Proc. IEEE Int. Conf. Rehabil. Robot.*, Jun. 2011, p. 6.
- [35] A. Rampp, J. Barth, S. Schülein, K.-G. Gaßmann, J. Klucken, and B. M. Eskofier, "Inertial sensor-based stride parameter calculation from gait sequences in geriatric patients," *IEEE Trans. Biomed. Eng.*, vol. 62, no. 4, pp. 1089–1097, Apr. 2015.
- [36] T. Watanabe, H. Saito, E. Koike, and K. Nitta, "A preliminary test of measurement of joint angles and stride length with wireless inertial sensors for wearable gait evaluation system," *Comput. Intell. Neurosci.*, vol. 2011, pp. 1–12, Jan. 2011.
- [37] S. Zhu, H. Anderson, and Y. Wang, "A real-time on-chip algorithm for IMU-based gait measurement," in *Proc. Pacific-Rim Conf. Multimedia*, in Lecture Notes in Computer Science, vol. 7674, 2012, pp. 93–104.
- [38] D. McGrath, B. R. Greene, K. J. O'Donovan, and B. Caulfield, "Gyroscope-based assessment of temporal gait parameters during treadmill walking and running," *Sports Eng.*, vol. 15, no. 4, pp. 207–213, Dec. 2012.
- [39] F. Bugané *et al.*, "Estimation of spatial-temporal gait parameters in level walking based on a single accelerometer: Validation on normal subjects by standard gait analysis," *Comput. Methods Programs Biomed.*, vol. 108, no. 1, pp. 129–137, 2012.
- [40] H. Rouhani, J. Favre, X. Crevoisier, and K. Aminian, "Measurement of multi-segment foot joint angles during gait using a wearable system," *J. Biomech. Eng.*, vol. 134, no. 6, p. 61006, Jun. 2012.
- [41] A. Dalton, H. Khalil, M. Busse, A. Rosser, R. van Deursen, and G. Ólaighin, "Analysis of gait and balance through a single triaxial accelerometer in presymptomatic and symptomatic Huntington's disease," *Gait Posture*, vol. 37, no. 1, pp. 49–54, Jan. 2013.
- [42] O. Tirosh, R. Begg, E. Passmore, and N. Knopp-Steinberg, "Wearable textile sensor sock for gait analysis," in *Proc. 7th Int. Conf. Sens. Technol. (ICST)*, Dec. 2013, pp. 618–622.
- [43] J. C. van den Noort, A. Ferrari, A. G. Cutti, J. G. Becher, and J. Harlaar, "Gait analysis in children with cerebral palsy via inertial and magnetic sensors," *Med. Biol. Eng. Comput.*, vol. 51, no. 4, pp. 377–386, Apr. 2013.
- [44] B. Mariani, M. C. Jiménez, F. J. G. Vingerhoets, and K. Aminian, "On-shoe wearable sensors for gait and turning assessment of patients with Parkinson's disease," *IEEE Trans. Biomed. Eng.*, vol. 60, no. 1, pp. 155–158, Jan. 2013.
- [45] S. Tadano, R. Takeda, and H. Miyagawa, "Three dimensional gait analysis using wearable acceleration and gyro sensors based on quaternion calculations," *Sensors*, vol. 13, no. 7, pp. 9321–9343, Jul. 2013.
- [46] F. Martínez-Martí, M. S. Martínez-García, S. G. García-Díaz, J. Garcia-Jiménez, A. J. Palma, and M. A. Carvajal, "Embedded sensor insole for wireless measurement of gait parameters," *Australas. Phys. Eng. Sci. Med.*, vol. 37, no. 1, pp. 25–35, Mar. 2014.
- [47] C. Wong, R. M. Kwasnicki, B. Heller, G. A. Tew, and G.-Z. Yang, "Gait parameter estimation from a miniaturized ear-worn sensor using singular spectrum analysis and longest common subsequence," *IEEE Trans. Biomed. Eng.*, vol. 61, no. 4, pp. 1261–1273, Apr. 2014.
- [48] A. B. Bourgeois, B. Mariani, K. Aminian, P. Y. Zambelli, and C. J. Newman, "Spatio-temporal gait analysis in children with cerebral palsy using, foot-worn inertial sensors," *Gait Posture*, vol. 39, no. 1, pp. 436–442, Jan. 2014.
- [49] A. Godfrey, S. Del Din, G. Barry, J. C. Mathers, and L. Rochester, "Within trial validation and reliability of a single tri-axial accelerometer for gait assessment," in *Proc. 36th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Aug. 2014, pp. 5892–5895.
- [50] T. Liu, Y. Inoue, K. Shibata, K. Shiojima, and M. M. Han, "Triaxial joint moment estimation using a wearable three-dimensional gait analysis system," *Measurement*, vol. 47, pp. 125–129, Jan. 2014.
- [51] Y. Mengüç et al., "Wearable soft sensing suit for human gait measurement," Int. J. Robot. Res., vol. 33, no. 14, pp. 1748–1764, Dec. 2014.
- [52] K. B. Mansour, N. Rezzoug, and P. Gorce, "Analysis of several methods and inertial sensors locations to assess gait parameters in able-bodied subjects," *Gait Posture*, vol. 42, no. 4, pp. 409–414, Oct. 2015.
- [53] C. M. Kanzler *et al.*, "Inertial sensor based and shoe size independent gait analysis including heel and toe clearance estimation," in *Proc. 37th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Aug. 2015, pp. 5424–5427.

- [54] S. Crea, C. Cipriani, M. Donati, M. C. Carrozza, and N. Vitiello, "Providing time-discrete gait information by wearable feedback apparatus for lower-limb amputees: Usability and functional validation," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 23, no. 2, pp. 7–250, Mar. 2015.
- [55] S. Shirai, I. Yabe, M. Matsushima, Y. M. Ito, M. Yoneyama, and H. Sasaki, "Quantitative evaluation of gait ataxia by accelerometers," *J. Neurol. Sci.*, vol. 358, nos. 1–2, pp. 253–258, Nov. 2015.
- [56] C. Buckley, B. Galna, L. Rochester, and C. Mazza, "Attenuation of upper body accelerations during gait: Piloting an innovative assessment tool for Parkinson's disease," *BioMed Res. Int.*, vol. 2015, pp. 1–6, Oct. 2015.
- [57] H.-C. Chang, Y.-L. Hsu, S.-C. Yang, J.-C. Lin, and Z.-H. Wu, "A wearable inertial measurement system with complementary filter for gait analysis of patients with stroke or Parkinson's disease," *IEEE Access*, vol. 4, pp. 8442–8453, 2017.
- [58] S. Byun, J. W. Han, T. H. Kim, and K. W. Kim, "Test-retest reliability and concurrent validity of a single tri-axial accelerometer-based gait analysis in older adults with normal cognition," *PLoS ONE*, vol. 11, no. 7, Jul. 2016, Art. no. e0158956.
- [59] D. Jarchi, B. Lo, C. Wong, E. Ieong, D. Nathwani, and G.-Z. Yang, "Gait analysis from a single ear-worn sensor: Reliability and clinical evaluation for orthopaedic patients," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 24, no. 8, pp. 882–892, Aug. 2016.
- [60] W. Chen, Y. Xu, J. Wang, and J. Zhang, "Kinematic analysis of human gait based on wearable sensor system for gait rehabilitation," *J. Med. Biol. Eng.*, vol. 36, no. 6, pp. 843–856, 2016.
- [61] C. Soaz and K. Diepold, "Step detection and parameterization for gait assessment using a single Waist-Worn accelerometer," *IEEE Trans. Biomed. Eng.*, vol. 63, no. 5, pp. 933–942, May 2016.
- [62] L. Donath, O. Faude, E. Lichtenstein, G. Pagenstert, C. Nüesch, and A. Mündermann, "Mobile inertial sensor based gait analysis: Validity and reliability of spatiotemporal gait characteristics in healthy seniors," *Gait Posture*, vol. 49, pp. 371–374, Sep. 2016.
- [63] A. Pacilli et al., "A wearable setup for auditory cued gait analysis in patients with Parkinson's disease," in Proc. IEEE Int. Symp. Med. Meas. Appl. (MeMeA), May 2016, pp. 551–556.
- [64] N. Carbonaro, F. Lorussi, and A. Tognetti, "Assessment of a smart sensing shoe for gait phase detection in level walking," *Electronics*, vol. 5, no. 4, p. 78, Nov. 2016.
- [65] C. Tunca, N. Pehlivan, N. Ak, B. Arnrich, G. Salur, and C. Ersoy, "Inertial sensor-based robust gait analysis in non-hospital settings for neurological disorders," *Sensors*, vol. 17, no. 4, p. 825, Apr. 2017.
- [66] V. Agostini, L. Gastaldi, V. Rosso, M. Knaflitz, and S. Tadano, "A wearable magneto-inertial system for gait analysis (H-gait): Validation on normal weight and overweight/obese young healthy adults," *Sensors*, vol. 17, no. 10, p. 2406, Oct. 2017.
- [67] A. Suppa et al., "L-DOPA and freezing of gait in Parkinson's disease: Objective assessment through a wearable wireless system," Frontiers Neurol., vol. 8, p. 406, Aug. 2017.
- [68] S. Qiu, Z. Wang, H. Zhao, and H. Hu, "Heterogeneous data fusion for three-dimensional gait analysis using wearable MARG sensors," *Int. J. Comput. Sci. Eng.*, vol. 14, no. 3, pp. 222–233, 2017.
- [69] M. Boutaayamou *et al.*, "Algorithm for temporal gait analysis using wireless foot-mounted accelerometers," *Commun. Comput. Inf. Sci.*, vol. 690, pp. 236–254, Feb. 2017.
- [70] Y. Moon *et al.*, "Monitoring gait in multiple sclerosis with novel wearable motion sensors," *PLoS ONE*, vol. 12, no. 2, 2017, Art. no. e0171346.
- [71] K. Orlowski, F. Eckardt, F. Herold, N. Aye, J. Edelmann-Nusser, and K. Witte, "Examination of the reliability of an inertial sensor-based gait analysis system," *Biomed. Tech.*, vol. 62, no. 6, pp. 615–622, Nov. 2017.
- [72] S. A. Moore, A. Hickey, S. Lord, S. Del Din, A. Godfrey, and L. Rochester, "Comprehensive measurement of stroke gait characteristics with a single accelerometer in the laboratory and community: A feasibility, validity and reliability study," *J. Neuroeng. Rehabil.*, vol. 14, no. 1, pp. 1–10, Dec. 2017.
- [73] X. Chen, S. Liao, S. Cao, D. Wu, and X. Zhang, "An acceleration-based gait assessment method for children with cerebral palsy," *Sensors*, vol. 17, no. 5, p. 1002, May 2017.
- [74] S. Rogan, R. de Bie, and E. D. de Bruin, "Sensor-based foot-mounted wearable system and pressure sensitive gait analysis," *Zeitschrift Gerontol. Geriatrie*, vol. 50, no. 6, pp. 488–497, Aug. 2017.
- [75] J. L. Lanovaz, A. R. Oates, T. T. Treen, J. Unger, and K. E. Musselman, "Validation of a commercial inertial sensor system for spatiotemporal gait measurements in children," *Gait Posture*, vol. 51, pp. 14–19, Jan. 2017.

- [76] S. Khandelwal and N. Wickström, "Evaluation of the performance of accelerometer-based gait event detection algorithms in different realworld scenarios using the MAREA gait database," *Gait Posture*, vol. 51, pp. 84–90, Jan. 2017.
- [77] G. P. Panebianco, M. C. Bisi, R. Stagni, and S. Fantozzi, "Analysis of the performance of 17 algorithms from a systematic review: Influence of sensor position, analysed variable and computational approach in gait timing estimation from IMU measurements," *Gait Posture*, vol. 66, pp. 76–82, Oct. 2018.
- [78] R. M. Ardle *et al.*, "Gait in mild Alzheimer's disease: Feasibility of multi-center measurement in the clinic and home with body-worn sensors: A pilot study," *J. Alzheimer's Dis.*, vol. 63, no. 4, p. 1557, May 2018.
- [79] H. De Vroey *et al.*, "The implementation of inertial sensors for the assessment of temporal parameters of gait in the knee arthroplasty population," *Clin. Biomech.*, vol. 54, pp. 22–27, May 2018.
- [80] C. Caramia *et al.*, "IMU-based classification of Parkinson's disease from gait: A sensitivity analysis on sensor location and feature selection," *IEEE J. Biomed. Health Informat.*, vol. 22, no. 6, pp. 1765–1774, Nov. 2018.
- [81] S. Qiu, L. Liu, H. Zhao, Z. Wang, and Y. Jiang, "MEMS inertial sensors based gait analysis for rehabilitation assessment via multisensor fusion," *Micromachines*, vol. 9, no. 9, p. 442, Sep. 2018.
- [82] W. Teufl, M. Lorenz, M. Miezal, B. Taetz, M. Fröhlich, and G. Bleser, "Towards inertial sensor based mobile gait analysis: Eventdetection and spatio-temporal parameters," *Sensors*, vol. 19, no. 1, p. 38, Dec. 2018.
- [83] M. Fusca, F. Negrini, P. Perego, L. Magoni, F. Molteni, and G. Andreoni, "Validation of a wearable IMU system for gait analysis: Protocol and application to a new system," *Appl. Sci.*, vol. 8, no. 7, p. 1167, Jul. 2018.
- [84] N. Roth, C. F. Martindale, B. M. Eskofier, H. Gaßner, Z. Kohl, and J. Klucken, "Synchronized sensor insoles for clinical gait analysis in home-monitoring applications," *Current Directions Biomed. Eng.*, vol. 4, no. 1, pp. 433–437, Sep. 2018.
- [85] E. Papi, Y. N. Bo, and A. H. McGregor, "A flexible wearable sensor for knee flexion assessment during gait," *Gait Posture*, vol. 62, pp. 480–483, May 2018.
- [86] A. R. Anwary, H. Yu, and M. Vassallo, "Optimal foot location for placing wearable IMU sensors and automatic feature extraction for gait analysis," *IEEE Sensors J.*, vol. 18, no. 6, pp. 2555–2567, Mar. 2018.
- [87] A. M. Keppler *et al.*, "Validity of accelerometry in step detection and gait speed measurement in orthogeriatric patients," *PLoS ONE*, vol. 14, no. 8, Aug. 2019, Art. no. e0221732.
- [88] S. Qiu et al., "Body sensor network-based gait quality assessment for clinical decision-support via multi-sensor fusion," *IEEE Access*, vol. 7, pp. 59884–59894, 2019.
- [89] S. Qiu *et al.*, "Body sensor network-based robust gait analysis: Toward clinical and at home use," *IEEE Sensors J.*, vol. 19, no. 19, pp. 8393–8401, Oct. 2019.
- [90] P. Pierleoni *et al.*, "Validation of a gait analysis algorithm for wearable sensors," in *Proc. Int. Conf. Sens. Instrum. IoT Era (ISSI)*, Aug. 2019, pp. 1–6.
- [91] F. García-Pinillos *et al.*, "Agreement between the spatiotemporal gait parameters from two different wearable devices and highspeed video analysis," *PLoS ONE*, vol. 14, no. 9, Sep. 2019, Art. no. e0222872.
- [92] L. Meng, U. Martinez-Hernandez, C. Childs, A. A. Dehghani-Sanij, and A. Buis, "A practical gait feedback method based on wearable inertial sensors for a drop foot assistance device," *IEEE Sensors J.*, vol. 19, no. 24, pp. 12235–12243, Dec. 2019.
- [93] N. Lefeber, M. Degelaen, C. Truyers, I. Safin, and D. Beckwee, "Validity and reproducibility of inertial physilog sensors for spatiotemporal gait analysis in patients with stroke," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 27, no. 9, pp. 1865–1874, Sep. 2019.
- [94] C. Wang, Y. Kim, H. Shin, and S. D. Min, "Preliminary clinical application of textile insole sensor for hemiparetic gait pattern analysis," *Sensors*, vol. 19, no. 18, p. 3950, Sep. 2019.
- [95] R. De Ridder, J. Lebleu, T. Willems, C. De Blaiser, C. Detrembleur, and P. Roosen, "Concurrent validity of a commercial wireless trunk triaxial accelerometer system for gait analysis," *J. Sport Rehabil.*, vol. 28, no. 6, Aug. 2019.
- [96] J. Yang *et al.*, "Machine learning based adaptive gait phase estimation using inertial measurement sensors," in *Proc. Design Med. Devices Conf.*, Apr. 2019, pp. 12235–12243.

- [97] D. Phan, N. Nguyen, P. N. Pathirana, M. Horne, L. Power, and D. Szmulewicz, "A random forest approach for quantifying gait ataxia with truncal and peripheral measurements using multiple wearable sensors," *IEEE Sensors J.*, vol. 20, no. 2, pp. 723–734, Jan. 2020.
- [98] M. Lueken, L. Mueller, M. G. Decker, C. Bollheimer, S. Leonhardt, and C. Ngo, "Evaluation and application of a customizable wireless platform: A body sensor network for unobtrusive gait analysis in everyday life," *Sensors*, vol. 20, no. 24, p. 7325, Dec. 2020.
- [99] N. Muthukrishnan, J. J. Abbas, and N. Krishnamurthi, "A wearable sensor system to measure step-based gait parameters for Parkinson's disease rehabilitation," *Sensors*, vol. 20, no. 22, p. 6417, Nov. 2020.
- [100] N. Mani, P. Haridoss, and B. George, "A wearable ultrasonic-based ankle angle and toe clearance sensing system for gait analysis," *IEEE Sensors J.*, vol. 21, no. 6, pp. 8593–8603, Mar. 2021.
- [101] S. Qiu *et al.*, "Towards wearable-inertial-sensor-based gait posture evaluation for subjects with unbalanced gaits," *Sensors*, vol. 20, no. 4, p. 1193, Feb. 2020.
- [102] Y. Hutabarat, D. Owaki, and M. Hayashibe, "Quantitative gait assessment with feature-rich diversity using two IMU sensors," *IEEE Trans. Med. Robot. Bionics*, vol. 2, no. 4, pp. 639–648, Nov. 2020.
- [103] F. Amitrano *et al.*, "Design and validation of an E-textile-based wearable sock for remote gait and postural assessment," *Sensors*, vol. 20, no. 22, p. 6691, Nov. 2020.
- [104] J. Figueiredo, S. P. Carvalho, J. P. Vilas-Boas, L. M. Gonçalves, J. C. Moreno, and C. P. Santos, "Wearable inertial sensor system towards daily human kinematic gait analysis: Benchmarking analysis to MVN BIOMECH," *Sensors*, vol. 20, no. 8, p. 2185, Apr. 2020.
- [105] D. Renggli et al., "Wearable inertial measurement units for assessing gait in real-world environments," *Frontiers Physiol.*, vol. 11, p. 90, Feb. 2020.
- [106] L. Zhou et al., "Validation of an IMU gait analysis algorithm for gait monitoring in daily life situations," in Proc. 42nd Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC), Jul. 2020, pp. 4229–4232.
- [107] I. A. Vajs, V. N. Bobić, M. D. Đurić-Jovičić, and M. M. Janković, "Open-source application for real-time gait analysis using inertial sensors," in *Proc. 28th Telecommun. Forum (TELFOR)*, 2020, pp. 1–4.
- [108] H. Zhao, Z. Wang, S. Qiu, J. Li, F. Gao, and J. Wang, "Evaluation of inertial sensor configurations for wearable gait analysis," in *Big Data, Cloud Computing, and Data Science Engineering* (Studies in Computational Intelligence), vol. 844. Cham, Switzerland: Springer, 2020, doi: 10.1007/978-3-030-24405-7_13.
- [109] S. Vítecková, H. Horáková, K. Poláková, R. Krupicka, E. Ruzicka, and H. Brozová, "Agreement between the GAITRite system and the wearable sensor BTS G-walk for measurement of gait parameters in healthy adults and Parkinson's disease patients," *PeerJ*, vol. 8, p. e8835, May 2020.
- [110] C. Pradeau, N. Sturbois-Nachef, and E. Allart, "Concurrent validity of the ZeroWire footswitch system for the measurement of temporal gait parameters," *Gait Posture*, vol. 82, pp. 133–137, Oct. 2020.
- [111] L.-F. Shi, C.-X. Qiu, D.-J. Xin, and G.-X. Liu, "Gait recognition via random forests based on wearable inertial measurement unit," *J. Ambient Intell. Humanized Comput.*, vol. 11, no. 11, pp. 5329–5340, Nov. 2020.
- [112] K. Aminian, F. Dadashi, B. Mariani, C. Lenoble-Hoskovec, B. Santos-Eggimann, and C. J. Büla, "Gait analysis using shoe-worn inertial sensors: How is foot clearance related to walking speed?" in *Proc. ACM Int. Joint Conf. Pervas. Ubiquitous Comput.*, Sep. 2014, pp. 481–485.
- [113] S. Biesmans and P. Markopoulos, "Design and evaluation of SONIS, a wearable biofeedback system for gait retraining," *Multimodal Technol. Interact.*, vol. 4, no. 3, pp. 1–13, 2020.
- [114] K. R. Sheerin, D. Reid, D. Taylor, and T. F. Besier, "The effectiveness of real-time haptic feedback gait retraining for reducing resultant tibial acceleration with runners," *Phys. Therapy Sport*, vol. 43, pp. 173–180, May 2020.
- [115] S. Studenski, "Gait speed and survival in older adults," J. Amer. Med. Assoc., vol. 305, no. 1, p. 50, Jan. 2011.
- [116] M. Sharifi Renani, C. A. Myers, R. Zandie, M. H. Mahoor, B. S. Davidson, and C. W. Clary, "Deep learning in gait parameter prediction for OA and TKA patients wearing IMU sensors," *Sensors*, vol. 20, no. 19, p. 5553, Sep. 2020.
- [117] R. de Souza Baptista, A. P. L. Bo, and M. Hayashibe, "Automatic human movement assessment with switching linear dynamic system: Motion segmentation and motor performance," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 25, no. 6, pp. 628–640, Jun. 2017.

- [118] D. Owaki, Y. Sekiguchi, K. Honda, A. Ishiguro, and S.-I. Izumi, "Shortterm effect of prosthesis transforming sensory modalities on walking in stroke patients with hemiparesis," *Neural Plasticity*, vol. 2016, pp. 1–9, Jan. 2016.
- [119] A. Tura, L. Rocchi, M. Raggi, A. G. Cutti, and L. Chiari, "Recommended number of strides for automatic assessment of gait symmetry and regularity in above-knee amputees by means of accelerometry and autocorrelation analysis," *J. Neuroeng. Rehabil.*, vol. 9, no. 1, pp. 1–8, Feb. 2012.
- [120] V. Belluscio, E. Bergamini, M. Tramontano, R. Formisano, M. G. Buzzi, and G. Vannozzi, "Does curved walking sharpen the assessment of gait disorders? An instrumented approach based on wearable inertial sensors," *Sensors*, vol. 20, no. 18, p. 5244, Sep. 2020.
- [121] L. H. Sloot, M. M. van der Krogt, and J. Harlaar, "Self-paced versus fixed speed treadmill walking," *Gait Posture*, vol. 39, no. 1, pp. 478–484, 2014.
- [122] M. D. Chang, S. Shaikh, and T. Chau, "Effect of treadmill walking on the stride interval dynamics of human gait," *Gait Posture*, vol. 30, no. 4, pp. 431–435, Nov. 2009.



Dai Owaki (Member, IEEE) received the Ph.D. degree from the Department of Electrical and Communication Engineering, Graduate School of Engineering, Tohoku University, in 2009. From April 2009 to March 2011, he was an Assistant Professor with the Graduate School of Engineering, Tohoku University. From April 2011 to September 2017, he was an Assistant Professor with the Research Institute of Electrical Communication, Tohoku University. From October 2017 to March 2019, he was an Assistant Professor with

the Department of Robotics, Graduate School of Engineering, Tohoku University. Since April 2019, he has been an Associate Professor with the Department of Robotics, Graduate School of Engineering, Tohoku University. His main research interests are neuro-robotics, synthetic neuro-rehabilitation, and bio-hybrid systems. He received the 2008 IEEE Robotics and Automation Society Japan Chapter Young Award (ICRA 2008), the SICE Annual Conference Young Author's Award (2008), and the Young Scientists' Award, MEXT (2020).



Mitsuhiro Hayashibe (Senior Member, IEEE) received the B.S. degree from the Tokyo Institute of Technology in 1999 and the M.S. and Ph.D. degrees from the Graduate School of Engineering, The University of Tokyo, Japan, in 2001 and 2005, respectively. He was an Assistant Professor with the Department of Medicine, The Jikei University School of Medicine, from 2001 to 2006. He was a Postdoctoral Fellow with the Institute of National de Recherche en Informatique et en Automatique (INRIA), and the Laboratoire



Yonatan Hutabarat received the B.E. degree from the Department of Engineering Physics and Nuclear Engineering, Universitas Gadjah Mada, Indonesia, in 2015, and the M.S. degree from the Sirindhorn International Institute of Technology, Thammasat University, Thailand, in 2018. He is currently pursuing the Ph.D. degree with the Neuro-Robotics Laboratory, Graduate School of Biomedical Engineering, Tohoku University, Japan. His current research interests are gait analysis, wearable sensors, and machine learning. d'Informatique, de Robotique et de Microelectronique de Montpellier, CNRS/University of Montpellier, France, in 2007. Since 2008, he has been a tenured Research Scientist with INRIA and the University of Montpellier. He has been a Visiting Researcher with the RIKEN Brain Science Institute and TOYOTA Collaboration Center, since 2012. Since 2017, he has been a Professor with the Department of Robotics, Tohoku University, and the Founder of the Neuro-Robotics Laboratory. He is a Senior Member of the IEEE Engineering in Medicine and Biology Society. He was a recipient of the 15th Annual Delsys Prize 2017 for the innovation in electromyography from the De Luca Foundation, USA. He serves as the Technical Activity Board Member of the International Foundation of Robotics Research (IFRR).